Author’s response to reviews

Title: Migraine and Increased Risk of Developing Open Angle Glaucoma: A Population-Based Cohort Study

Authors:

Jehn-Yu Huang (d03421102@ntu.edu.tw)
Chien-Chia Su (kido0214@yahoo.com.tw)
Tsing-Hong Wang (q001421008@ntu.edu.tw)
I-Ju Tsai (hunch0815@hotmail.com)

Version: 4 Date: 29 Jan 2019

Author’s response to reviews:

January 29, 2019

Guangde Tu
Editor
BMC Ophthalmology


Dear Editor-in-Chief:

I, along with my coauthors, would like to re-submit the attached manuscript entitled “Migraine and Increased Risk of Developing Open Angle Glaucoma: A Population-Based Cohort Study” as an Original Research article. The manuscript ID is MS No.: BOPH-D-18-00734R3.
The manuscript has been carefully rechecked and appropriate changes have been made in accordance with the reviewers’ comments and the editorial policies. We have amended the Ethics approval and consent to participate section of the manuscript to address the relevant clarifications. All authors have reviewed and concur with each of the changes made to the manuscript, and to the order their names listed in the authorship.

We hope that the revised manuscript is now suitable for publication in your journal.

I look forward to your reply

Sincerely,

Chien-Chia Su, MD
Department of Ophthalmology, National Taiwan University Hospital
12F, No. 7, Zhongshan S. Rd. Taipei City, 100, Taiwan, R.O.C.
Tel.: 886-2-2356-5906 Fax: 886-2-2393-4420
E-mail: chienchiasu@ntu.edu.tw

Comments from the editor:

1. Thank you for responding to our requests and providing the relevant clarifications. However, before we can proceed, we require all of the clarifications you have provided in the Ethics approval and consent to participate section of your manuscript.

We appreciate the editor’s comment. We have amended the Ethics approval and consent to participate section of the manuscript to make the relevant clarifications.
Ethics approval: This research protocol has been reviewed and approved by the Research Ethics Committee of the National Taiwan University Hospital and in compliance with the categories of exempt in accordance with regulations of our hospital and National Health Research Institutes.

Consent to participate: The need for informed consent was waived by the IRB.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

We have uploaded the files according to the instructions. Thank you.